

### Smith ScholarWorks

**Biological Sciences: Faculty Publications** 

**Biological Sciences** 

12-9-2008

## The Impact of Repeated Rounds of Mass Drug Administration with Diethylcarbamazine Plus Albendazole on Bancroftian Filariasis in Papua New Guinea

Gary J. Weil Washington University School of Medicine

Will Kastens Case Western Reserve University School of Medicine

Melinda Susapu Papua New Guinea Institute of Medical Research

Sandra J. Laney Smith College

Steven A. Williams Smith College, swilliam@smith.edu

See next page for additional authors

Follow this and additional works at: https://scholarworks.smith.edu/bio\_facpubs

Part of the <u>Biology Commons</u>

### **Recommended** Citation

Weil, Gary J.; Kastens, Will; Susapu, Melinda; Laney, Sandra J.; Williams, Steven A.; King, Christopher L.; Kazura, James W.; and Bockarie, Moses J., "The Impact of Repeated Rounds of Mass Drug Administration with Diethylcarbamazine Plus Albendazole on Bancroftian Filariasis in Papua New Guinea" (2008). Biological Sciences: Faculty Publications, Smith College, Northampton, MA. https://scholarworks.smith.edu/bio\_facpubs/19

This Article has been accepted for inclusion in Biological Sciences: Faculty Publications by an authorized administrator of Smith ScholarWorks. For more information, please contact scholarworks@smith.edu

#### Authors

Gary J. Weil, Will Kastens, Melinda Susapu, Sandra J. Laney, Steven A. Williams, Christopher L. King, James W. Kazura, and Moses J. Bockarie

## The Impact of Repeated Rounds of Mass Drug Administration with Diethylcarbamazine Plus Albendazole on Bancroftian Filariasis in Papua New Guinea

# Gary J. Weil<sup>1</sup>\*, Will Kastens<sup>2</sup>, Melinda Susapu<sup>3</sup>, Sandra J. Laney<sup>4</sup>, Steven A. Williams<sup>4</sup>, Christopher L. King<sup>2</sup>, James W. Kazura<sup>2</sup>, Moses J. Bockarie<sup>2,3¤</sup>

1 Washington University School of Medicine, St. Louis, Missouri, United States of America, 2 Case Western Reserve University School of Medicine, Cleveland, Ohio, United States of America, 3 Papua New Guinea Institute of Medical Research, Madang, Papua New Guinea, 4 Smith College, Northampton, Massachusetts, United States of America

#### Abstract

**Background:** This study employed various monitoring methods to assess the impact of repeated rounds of mass drug administration (MDA) on bancroftian filariasis in Papua New Guinea, which has the largest filariasis problem in the Pacific region.

*Methodology/Principal Findings:* Residents of rural villages near Madang were studied prior to and one year after each of three rounds of MDA with diethylcarbamazine plus albendazole administered per World Health Organization (WHO) guidelines. The mean MDA compliance rate was 72.9%. Three rounds of MDA decreased microfilaremia rates (Mf, 1 ml night blood by filter) from 18.6% pre-MDA to 1.3% after the third MDA (a 94% decrease). Mf clearance rates in infected persons were 71%, 90.7%, and 98.1% after 1, 2, and 3 rounds of MDA. Rates of filarial antigenemia assessed by card test (a marker for adult worm infection) decreased from 47.5% to 17.1% (a 64% decrease) after 3 rounds of MDA. The filarial antibody rate (lgG<sub>4</sub> antibodies to Bm14, an indicator of filarial infection status and/or exposure to mosquito-borne infective larvae) decreased from 59.3% to 25.1% (a 54.6% decrease). Mf, antigen, and antibody rates decreased more rapidly in children <11 years of age (by 100%, 84.2%, and 76.8%, respectively) relative to older individuals, perhaps reflecting their lighter infections and shorter durations of exposure/infection prior to MDA. Incidence rates for microfilaremia, filarial antigenemia, and antifilarial antibodies also decreased significantly after MDA. Filarial DNA rates in *Anopheles punctulatus* mosquitoes that had recently taken a blood meal decreased from 15.1% to 1.0% (a 92.3% decrease).

**Conclusions/Significance:** MDA had dramatic effects on all filariasis parameters in the study area and also reduced incidence rates. Follow-up studies will be needed to determine whether residual infection rates in residents of these villages are sufficient to support sustained transmission by the *An. punctulatus* vector. Lymphatic filariasis elimination should be feasible in Papua New Guinea if MDA can be effectively delivered to endemic populations.

Citation: Weil GJ, Kastens W, Susapu M, Laney SJ, Williams SA, et al. (2008) The Impact of Repeated Rounds of Mass Drug Administration with Diethylcarbamazine Plus Albendazole on Bancroftian Filariasis in Papua New Guinea. PLoS Negl Trop Dis 2(12): e344. doi:10.1371/journal.pntd.0000344

Editor: Patrick J. Lammie, Centers for Disease Control and Prevention, United States of America

Received August 4, 2008; Accepted November 11, 2008; Published December 9, 2008

**Copyright:** © 2008 Weil et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This collaborative work was supported by NIH grants Al06517, Al033061, Al03585 and Al065715. The funders had no role in study design, data collection, decision to publish, or preparation of the manuscript.

**Competing Interests:** The filariasis antigen card test used in this study uses reagents licensed from Barnes-Jewish Hospital, an affilation of Gary J. Weil. All royalties from the sales of this test are donated to the Barnes-Jewish Hospital Foundation, a registered not-for-profit organization. This statement is made in the interest of full disclosure, and not because the authors consider this to be a conflict of interest.

\* E-mail: GWEIL@im.wustl.edu

¤ Current address: Centre for Neglected Tropical Diseases, Liverpool School of Tropical Medicine, Liverpool, United Kingdom

#### Introduction

Lymphatic filariasis (LF) is a deforming and disabling infectious disease that causes elephantiasis and hydroceles. The infection affects some 120 million people in an estimated 83 countries in tropical and subtropical regions, with an estimated 1.2 billion individuals at risk [1]. Most LF is caused by *Wuchereria bancrofti*, a nematode parasite that is transmitted to humans by mosquitoes. The World Health Assembly passed a resolution in 1997 that called for global elimination of LF as a public health problem

(WHA Resolution 50.29, see www.filariasis.org). The World Health Organization (WHO) developed a plan for elimination that is based on selective diagnosis to identify endemic areas followed by repeated, annual cycles of mass drug administration (MDA) of antifilarial medications [1,2]. The most recent summary from WHO reported that approximately 1.9 billion doses of MDA were distributed to more than 500 million individuals between 2000 and 2007 [3]. Thus, the Global Programme to Eliminate Lymphatic Filariasis (GPELF) is the largest infectious disease intervention program attempted to date based on MDA. Applied

#### **Author Summary**

Lymphatic filariasis (LF) is a deforming and disabling disease that is caused by parasitic worms that are transmitted by mosquitoes. While a number of countries have initiated LF elimination programs based on mass drug administration (MDA), relatively little good information is available on the impact of MDA on filariasis prevalence and incidence rates in populations. This study assessed the impact of three rounds of MDA (with single doses of diethylcarbamazine and albendazole) on filariasis infection rates in villages in Papua New Guinea, which has the largest filariasis problem in the Pacific region. MDA dramatically reduced rates for all filariasis infection markers tested. These included microfilaremia (parasites in blood that are necessary for transmission of the infection), filarial antigenemia (a marker for adult worm infection), antifilarial antibodies (which indicate infection or heavy exposure to the parasite), and parasites in mosquitoes that transmit the infection. In addition to curing existing infections, MDA also reduced new infection rates in the study population to very low levels. These results suggest that it should be possible to eliminate LF in Papua New Guinea if MDA can be effectively delivered to endemic populations.

field research is needed to validate the GPELF strategy and to test methods for measuring the impact of MDA. Recent papers have reported encouraging data on the impact of MDA with diethylcarbamazine (DEC) plus albendazole on various filariasis parameters in Egypt and emphasized the importance of compliance in MDA programs [4,5]. However, more information is needed from areas with different epidemiological and ecological parameters, e.g. mosquito vectorial capacity, transmission intensity, baseline infection rates, and climate.

Bancroftian filariasis is a major public health problem in Papua New Guinea [6]. With approximately 39% of its population infected, the country has some of the highest LF infection rates in the world [7], despite the fact the principal vectors there (Anopheles punctulatus group, which also transmit malaria) are less efficient for filariasis transmission than Culex or Aedes mosquitoes that serve as vectors in other endemic regions [6,8]. At this time, Papua New Guinea has more filariasis cases than any other country in the Pacific region [9]. Although it is the only country in the region that has not yet fully implemented a national LF elimination program, the Papua New Guinea Department of Health has recently initiated a MDA program for elimination in several provinces. In some areas, MDA will overlap with the distribution of insecticidetreated bednets for malaria control. This overlap in public health interventions may benefit LF elimination efforts, as suggested by reports from the Solomon Islands where DDT spraying (used for malaria control) reduced LF transmission [10].

Several studies conducted over the past 10 years have assessed the impact of MDA on LF infection parameters in Papua New Guinea. For example, a study performed in 14 villages in East Sepik Province found that four rounds of MDA with DEC plus ivermectin or with DEC alone (drug regimens that have been superseded by the current WHO recommendation of single dose DEC plus albendazole) dramatically decreased microfilaremia (Mf) prevalence rates and transmission parameters in areas with high and moderate baseline infection and transmission rates [11]. Other studies in offshore island communities in Papua New Guinea found dramatic decreases in Mf and/or filarial antigenemia rates following MDA or community distribution of DEC alone or DEC with albendazole [12,13]. We now report results from a community study that was designed to evaluate the efficacy of repeated rounds of MDA with DEC plus albendazole (distributed according to WHO guidelines [14]) on a number of filariasis infection parameters. Results of this study should be useful for those responsible for planning and implementing national LF elimination programs in Papua New Guinea and in other areas where filariasis is transmitted by *Anopheles* mosquitoes.

#### Methods

#### Study site

The study was performed in 4 villages in the Usino-Bundi district, Madang Province, Papua New Guinea. The villages are located approximately 40 km southwest of the provincial capital, Madang town. The study villages were located at least 2 km from untreated filariasis-endemic villages. This is the estimated flight range of the local mosquito vector [15]. Villages were mapped and censuses were conducted prior to the first round of MDA and annually thereafter. Two Infection prevalence surveys were performed before the first round of MDA, and this was repeated approximately 12 months after each of three annual rounds of MDA.

#### Field procedures and MDA

Field teams comprised of trained nurses and field technicians met with community leaders and held outdoor community education sessions to inform people about the health significance and biology of LF, the planned MDA program, and the importance of blood tests for monitoring the impact of MDA. Annual follow-up meetings were held to communicate preliminary results and to provide community members with opportunities to ask questions about the project.

Community liaison personnel mobilized village residents to participate in the study. Children <2 years of age, children who weighed <10 kg, pregnant women, and people with severe chronic illness or acute illness with fever were excluded from the study. We reassessed eligibility for treatment each year. For example, women who were pregnant in year 1 were eligible for the study in later years if they were not pregnant. Survey teams enrolled subjects in the late afternoon by obtaining oral informed consent from adults and recording demographic information. Enrollment of children required their assent and consent from at least one parent. The population was surveyed twice prior to the first round of MDA. The first pre-MDA survey (pre-MDA-A, which collected day blood samples for antigen and antibody testing only without microfilaria testing) was performed approximately 2 years prior to the first round of MDA. The second pre-MDA survey (pre-MDA-B, with blood for Mf, antigen, and antibody testing) was performed together with the first round of MDA (with blood collection just prior to treatment).

MDA comprised the WHO-recommended regimen of a single oral dose of diethylcarbamazine citrate (6 mg/kg body weight) with albendazole (400 mg regardless of weight). This was repeated once per year for a total of three years. Project staff directly observed ingestion of antifilarial medications. Some people agreed to receive MDA but refused to provide blood samples. MDA compliance rates were estimated by dividing the number of people who ingested DEC with albendazole by the number of village residents in the census who were at least 2 years of age. This is similar to "observed coverage" in GPELF guidelines [14]. However, persons ineligible for treatment (e.g., pregnant women) are included in the denominator for our compliance calculations but not in the GPELF "observed coverage".

#### Blood tests for filariasis infection or exposure

All tests were performed in the Papua New Guinea Institute of Medical Research Laboratories near Madang. Microfilaremia was assessed by membrane filtration of 1 ml of venous blood collected between 9 p.m. and 12 midnight [16]. The Mf prevalence rate was calculated by dividing the number of people who were Mf positive by the number of people tested for Mf. Few blood samples were collected from children younger than 6 years of age, and results from these samples are not included in this report. We also calculated community microfilaria load (CMFL) as a measure of the total amount of Mf present in study communities as previously described [17]. CMFL is defined as the antilog of the [sum of log (X+1)/N]–1, where X is the MF count in positive subjects and N is the number of people tested for microfilaremia.

Circulating filarial antigenemia (CFA) was assessed with a rapidformat antigen card test (Filariasis Now, Binax Inc., Portland, Maine, USA) [18,19]. Test instructions call for testing 100  $\mu$ l of whole blood. We tested 70  $\mu$ l of plasma, which is the approximate volume of plasma in 100  $\mu$ l of whole blood. Card tests were read visually at 10 minutes. IgG<sub>4</sub> antibodies to recombinant filarial antigen Bm14 in human plasma were detected by ELISA as previously described [20]. Sera were tested in duplicate, and borderline samples were retested. Prior studies have shown that this test is sensitive and specific for infection or heavy exposure to filarial parasites [20–22].

#### Molecular xenomonitoring

Mosquitoes were collected with CDC light traps (without dry ice) placed in houses (1 night per house) in study villages over 4 months period before MDA commenced and for 4 months starting 6 to 9 months following each round of MDA. Two pre-MDA mosquito collections were conducted. The first (pre-MDA-A) was conducted approximately 12 to 18 months prior to the first round of MDA; the second (pre-MDA-B) was 1 to 5 months prior to the first round of MDA.

Blood-engorged, gravid, or semi-gravid An. punctulatus group mosquitoes (all of which had recently ingested blood) were identified by morphology and sorted into pools by household. W. bancrofti DNA was detected in mosquito pools by PCR as previously described in detail [23–25], except mosquitoes were macerated by vortexing with ball bearings [26] instead of by mortar and pestle. Mosquito infection rates (maximum likelihood estimates with 95% confidence intervals) were calculated with PoolScreen 2.02 software [27].

#### Data entry and statistical methods

Data entry was performed with Visual FoxPro software with field limits and double data entry. We used the SPSS v.14 software package (SPSS, Chicago, IL) for statistical analysis. The chi-square test was used to assess the significance of differences in proportions.

#### Ethical clearance

This study (including the oral consent process) was reviewed and approved by institutional review boards at Washington University School of Medicine and Case Western Reserve University and the Medical Research Advisory Committee of the Papua New Guinea Department of Health. The sponsor of the study (Division of Microbiology and Infectious Diseases, NIAID, NIH, Bethesda, MD) also reviewed the study protocol to ensure compliance with GCP standards. Study personnel informed prospective study participants about the study by reading them a consent document in the local language. Oral consent was documented on case report forms. Participation by children required consent from at least one parent and the child's assent. As noted above, some subjects agreed to take anti-filarial medications but refused blood tests.

#### Results

#### Pre-MDA human infection parameters

Table 1 shows baseline infection indicators by age group just prior to the first round of MDA (pre-MDA-B survey data). Mf prevalence increased dramatically up to age 20 years, with slightly higher rates in older age groups. The mean±SD of the number of Mf per ml in 106 microfilaremic subjects was  $610\pm1073$  (median 161; range 1-5,960). CFA and Bm14-specific IgG<sub>4</sub> antibody rates increased dramatically from age 6-10 years to 21-30 years, with slight increases (CFA) or little change (Bm14) in older age groups. Antibody and CFA rates were much higher than Mf rates in all age groups. Persons with Mf or filarial antigenemia had a significantly higher antibody rate (81.9%, n = 279) than persons with negative Mf and CFA tests (37.6%, N = 287) (P<0.001). Antibody rates were higher than CFA rates in younger age groups  $(\leq 30$  years of age, P < 0.001), but CFA rates were similar to antibody rates in older people (difference not significant). Filarial infection rates (Mf and CFA) were significantly higher in males than in females (Table 2). Males also had higher antifilarial antibody rates, but the difference was not statistically significant. Gender differences in infection and antibody rates were much more striking in adults than in children 6 to 15 years of age, although the same trends were present in children.

## Effects of MDA on filariasis infection parameters in humans

The mean MDA compliance rate over the 3 rounds of MDA was 72.9% (Table 3). However, compliance rates were much lower in children 2–5 years of age compared to those in older individuals.

MDA had dramatic effects on all filariasis parameters examined (Table 4). Over the three annual rounds of MDA, microfilaremia (Mf prevalence rates and CMFL) decreased more rapidly and to lower levels than CFA or antibody rates. Antigenemia and antibody prevalence rates in young children decreased more after MDA than rates for the total study population.

Seven of 529 subjects tested had Mf one year after the third round of MDA. Mf counts in these subjects ranged from 1-52 Mf/ml (mean  $18.0\pm20.5$ ; median 9). These subjects tended to have

**Table 1.** Filariasis infection parameters by age in Usino villages just prior to the first round of mass drug administration.

Age Range	Mf (%)	CFA (%)	Bm14 (%)
6–10	3/125 (2.4)	22/120 (18.3)	43/120 (35.8)
11-15	7/90 (7.8)	25/87 (28.7)	40/87 (46.0)
16-20	13/60 (21.7)	27/60 (45)	37/60 (61.7)
21-30	31/131 (24.4)	78/128 (60.9)	95/129 (73.6)
31–40	24/80 (30.0)	52/79 (65.8)	54/79 (68.4)
41-50	24/49 (29.2)	34/47 (72.3)	37/48 (77.1)
51+	13/37 (35.1)	27/37 (73.0)	26/37 (70.3)
Total	106/571 (18.6)	265/558 (47.5)	332/560 (59.3)

Abbreviations: Mf, microfilaremia; CFA, circulating filarial antigenemia; Bm14, antibodies to recombinant filarial antigen Bm14.

Data shown are the number of positive tests and the total number of tests performed (% positive).

doi:10.1371/journal.pntd.0000344.t001

**Table 2.** Filariasis infection parameters by gender in Usino villages just prior to the first round of mass drug administration.

Gender	Mf Prev (%)	CFA Prev (%)	Bm14 (%)
Females 6–15	2/89 (2.2)	15/87 (17.2)	33/87 (37.9)
Males 6–15	8/126 (6.3)	32/120 (26.7)	50/120 (41.7)
P value	0.28	0.15	0.69
Females >15	32/182 (17.6)	99/181 (54.7)	117/181 (64.6)
Males >15	64/174 (36.8)	119/170 (70)	132/172 (76.7)
P value	<0.001	0.005	0.02
All Females >5	34/271 (12.5)	114/268 (42.5)	150/268 (56.0)
All Males >5	72/300 (24.0)	151/290 (52.1)	182/292 (62.3)
P value	0.001	0.024	0.126

Abbreviations: Mf, microfilaremia; CFA, circulating filarial antigenemia; Bm14, antibodies to recombinant filarial antigen Bm14.

Data shown are the number of positive tests and the total number of tests performed (% positive). Numbers in the first column refer to age in years. doi:10.1371/journal.pntd.0000344.t002

high Mf counts prior to MDA. Only one of the subjects with Mf detected after MDA round 3 had not received MDA; two had been treated once, one had been treated twice, and two had been treated three times.

#### Sensitivity of the CFA and Bm14 antibody tests in Mfpositive subjects before and after MDA

Table 5 shows CFA and antibody rates in Mf-positive subjects detected in study villages before MDA and after each round of MDA. These rates did not change significantly after MDA. *P* values for differences in CFA and Bm14 antibody rates (for samples from Mf-positive subjects) in different years of the study were 0.60 and 0.78, respectively (chi-square).

#### Longitudinal observations on microfilaremia after MDA

Longitudinal Mf data were available for 104 subjects who were initially Mf positive and who had repeat Mf testing approximately one year after one or more rounds of MDA until Mf clearance was achieved. Mf clearance rates were 76% (79/107), 95.2% (99/104), and 98.1% (102/104) after 1, 2, or 3 rounds of treatment, respectively; 2 subjects had persistent microfilaremia after 3 rounds of treatment. Subjects with higher baseline Mf counts tended to require more rounds of MDA to completely clear Mf. The geometric mean and median Mf counts for 79 people who cleared Mf after one round of MDA were 64.8 and 89 (range, 1–3813); geometric mean and median Mf counts for 25 subjects who failed to clear Mf after one round of MDA were 604 and 833 (range 11–

5279). The difference in baseline Mf counts between the two groups was statistically significant (P<0.001 by the Mann Whitney U test).

Microfilaria incidence rates in persons who were Mf-negative just prior to MDA were 1.7% (7/408), 0.4% (2/555), and 0.2% (1/472) after MDA rounds 1, 2, and 3, respectively. Note that Mf incidence data were not available for the period between the two pre-MDA surveys. Despite this, the decrease in Mf incidence rates after MDA was statistically significant (P = 0.013 by chi square). The mean Mf count in persons with Mf incidence was  $86.2\pm112$ Mf/ml (median 26, range 7-353). Of course, some of these cases may represent "pseudo-incidence" due to technical problems with Mf detection or clerical errors. Most people with Mf incidence had positive antigen (7/10) and antibody (8/10) tests one year prior to the appearance of Mf. Four subjects with microfilaremia at baseline were Mf-negative after MDA but positive again with counts ranging from 47 to 68 Mf/ml one year later. Such cases of temporary Mf clearance following treatment were not counted in clearance or incidence results shown above.

## Decreased antibody and antigenemia incidence rates after MDA

Table 6 shows Bm14 antibody and CFA incidence events and rates for the total study population and for children 6–15 years of age. Incidence rates for the 2 year period between the 2 pre-MDA surveys were high. Pre-MDA incidence rates in children <16 years of age were not significantly different from those in older people. Incidence rates decreased dramatically after MDA. Incidence rates for the 2 year period after the first round of MDA were significantly lower than those for the two year period between the pre-MDA-A and pre-MDA-B surveys (Table 6). Incidence rates for antifilarial antibodies and CFA decreased by 75% and 63% in the total study population (and by 68% and 70% in children), respectively, after 3 rounds of MDA.

#### Effect of MDA on filarial DNA rates in mosquitoes

573 pools of recently fed (blood fed, gravid, or semigravid) mosquitoes (3,729 mosquitoes) were tested for filarial DNA by PCR (Table 7). The mosquitoes were collected from  $114.6\pm16.9$  houses per year (approximately 70% of the houses in study villages). The mean number of recently fed mosquitoes per pool was  $6.5\pm6.6$  (median 4.0; range 1–28). The high baseline rate of filarial DNA in mosquitoes just prior to MDA (15.1% in the pre-MDA-B survey) decreased rapidly after MDA (Table 7). The 92.3 % decrease in the filarial DNA rate after 3 rounds of MDA was similar to the 93% decrease in the Mf prevalence rate observed in the human population.

#### Discussion

This is the first detailed report on the effects of repeated rounds of MDA with DEC and albendazole on filariasis infection

Table 3. Mass drug administration (MDA) compliance rates in Usino, Papua New Guinea.

Age (yr)	Round 1		Round 2		Round 3	
	Census	MDA Compliance %	Census	MDA Compliance %	Census	MDA Compliance %
2-5	147	29.9	146	39	130	23.1
>5	825	75.3	854	82.8	841	81.7
≥2	972	68.4	1000	76.4	971	73.9

doi:10.1371/journal.pntd.0000344.t003

Table 4. Filariasis parameters before and after mass drug administration (MDA) in Usino villages.

		Mf (%)	CMFL	CFA (%)	Bm14 (%)
All ages	Pre-MDA-A	n/a	n/a	334/627 (53.3)	411/683 (60.2)
	Pre-MDA-B	106/757 (18.6)	1.45	265/558 (47.5)	332/560 (59.3)
	MDA-1	58/696 (8.3)	0.44	243 692 (35.1)	276/696 (39.7)
	MDA-2	24/714 (3.4)	0.16	175/695 (25.2)	337/693 (48.8)
	MDA-3	7/529 (1.3)	0.03	93/543 (17.1)	138/550 (25.1)
	% Decrease	93.0	97.9	64.0	54.6
Age <11	Pre-MDA-A	n/a	n/a	26/130 (20.0)	52/137 (38.0)
	Pre-MDA-B	3/125 (2.4)		22/120 (18.3)	43/120 (35.8)
	MDA-1	1/138 (0.7)		14/137 (10.1)	17/138 (12.3)
	MDA-2	0/129 (0)		3/124 (2.4)	21/124 (16.3)
	MDA-3	0/129 (0)		3/105 (2.9)	9/109 (8.3)
	% Decrease	100		84.2	76.8

Data shown are the number of positive tests and the total number of tests performed (% positive).

The pre-MDA-A survey did not include Mf testing. CMFL was calculated for all ages only.

Pre-MDA-B values were used as the baseline for calculating % decrease for rates after MDA.

Abbreviations: Mf, microfilaremia; CMFL, community microfilarial load (see Methods); CFA, circulating filarial antigenemia; Bm14, antibodies to recombinant filarial antigen Bm14.

doi:10.1371/journal.pntd.0000344.t004

parameters in an area with *Anopheles* transmission. The study site had baseline filarial infection rates that were high in the global context but moderate for Papua New Guinea. Baseline infection rates were higher in older age groups and higher in males than in females. These trends probably reflect increased cumulative exposure in older people and differences in exposure between males and females. Since the gender differences were much more striking in adults, these could also be related to biological factors such as hormone levels.

The performance of the ICT antigen test and the Bm14 antibody test require some comment before we address the impact of MDA on filariasis parameters in this study. The ICT test detected filarial antigenemia in a high percentage of untreated Mf-positive subjects (detected by membrane filtration of 1 ml of venous blood), and this sensitivity was maintained after MDA. This is in contrast to a recent study from Kenya that reported decreased sensitivity of the ICT test in Mf carriers (detected by the counting chamber method with 0.1 ml of finger prick blood) after 2 rounds of MDA [28]. The Bm14 antibody test was less sensitive than the CFA test in Mf-positive subjects in the present study, and

**Table 5.** Filarial antigenemia and Bm14 antibody rates insubjects with microfilaremia before and after mass drugadministration (MDA).

CFA (%)	Bm14 (%)	
98/104 (94.2)	80/104 (76.9)	
56/58 (96.6)	41/58 (70.7)	
21/23 (91.3)	18/23 (78.3)	
6/7 (85.7)	6/7 (85.7)	
	98/104 (94.2) 56/58 (96.6) 21/23 (91.3)	98/104 (94.2)         80/104 (76.9)           56/58 (96.6)         41/58 (70.7)           21/23 (91.3)         18/23 (78.3)

Data shown are the number of subjects with positive tests and the number of Mf-positive subjects tested (% positive) before the first round of MDA and approximately 12 months after each round of MDA.

Abbreviations: Mf, microfilaremia; CFA, circulating filarial antigenemia; Bm14,  $IgG_4$  antibodies to recombinant filarial antigen Bm14.

doi:10.1371/journal.pntd.0000344.t005

it was also somewhat less sensitive than previously reported [20,22]. However, both the CFA test and the Bm14 antibody test were much more sensitive than Mf detection for detecting filariasis activity in the study communities, and this is consistent with prior reports [5,29]. Antibody rates in children <11 years of age were much higher than Mf or CFA rates, both before and after MDA. This supports the strategy of testing sentinel populations of young

**Table 6.** Incidence events and incidence rates for anti-filarialantibodies and filarial antigenemia before and after mass drugadministration (MDA).

	Survey	Bm14 Incidence (rate, %)	CFA Incidence (rate, %)
All ages			
	Pre-MDA-B	21/154 (6.8)	11/120 (4.6)
	MDA-1	5/164 (3.0)	2/201 (1.0)
	MDA-2	7/213 (3.3)	3/287 (1.0)
	MDA-3	3/181 (1.7)	5/291 (1.7)
Age <16			
	Pre-MDA-B	14/83 (8.4)	7/56 (6.3)
	MDA-1	4/98 (4.1)	1/118 (0.8)
	MDA-2	4/131 (3.1)	0/143 (0)
	MDA-3	3 /111 (2.7)	3/155 (1.9)

Incidence events for each test were defined as positive tests in people who had tested negative in the prior survey and who had never had a positive test in prior years. This analysis is restricted to persons with data available for consecutive years. The pre-MDA-B incidence data are for a 2 year interval (with annual incidence rates shown in parentheses). Other incidence data are for one year intervals. Annual incidence rates for the 2 year period between pre-MDA-A and pre-MDA-B were significantly higher than rates for the two years following the first round of MDA (P<0.001 by chi-square for all comparisons).

Abbreviations: CFA, circulating filarial antigenemia; Bm14,  $\rm IgG_4$  antibodies to recombinant filarial antigen Bm14.

doi:10.1371/journal.pntd.0000344.t006

Table 7. Mosquito collections and filarial DNA rates in mosquitoes before and after mass drug administration (MDA).

Survey	No. Pools	Mosq. per Pool (SD)	No. Mosquitoes Tested	Filarial DNA Rate (95% CI)
Pre-MDA-A	97	6.8 (6.1)	565	11.4 (10.87–21.25
Pre-MDA-B	119	7.5 (7.3)	892	15.1 (11.7–19.2)
MDA-1	118	6.2 (6.4)	726	3.7 (2.2–5.6)
MDA-2	130	6.7 (6.8)	926	4.8 (3.3–6.7)
MDA-3	100	6.2 (6.4)	620	1.02 (0.4–2.2)
Totals	573	6.5 (6.6)	3,729	

Filarial DNA was detected in mosquitoes by PCR, and rates (maximum likelihood with 95% CI) were estimated by PoolScreen. The filarial DNA rate in mosquitoes collected in the MDA-3 survey (6 to 9 months after the third round of MDA) represented a 92.3% decrease from the rate in mosquitoes collected during the pre-MDA-B survey (just prior to the first round of MDA).

doi:10.1371/journal.pntd.0000344.t007

children for antifilarial antibodies as a means of assessing recent filariasis activity in communities [30].

Population MDA compliance rates were very good throughout the 3-year study period. However, this required a lot of effort, with multiple visits to the study villages and labor-intensive recruitment of village residents. It might be easier to achieve high compliance rates in a national MDA program that was not linked to collection of venous blood. MDA compliance rates were low in children <6 years of age. This may have reflected parents' concerns about blood tests in their young children. We believe that the national LF elimination program in Papua New Guinea will need to develop new strategies to achieve high MDA compliance in young children. Information campaigns should emphasize the dual benefits of MDA on LF and soil-transmitted helminth infections.

MDA dramatically reduced all filariasis infection parameters in the study villages. As in earlier studies, Mf rates in people and parasite DNA rates in mosquitoes fell more rapidly than CFA or antibody rates [5,31,32]. While low residual filarial DNA rates in mosquitoes indicate the presence of Mf carriers in communities following MDA, this does not necessarily mean that significant LF transmission will continue in these areas. PCR can detect DNA from dead filarial parasites in mosquitoes [33], and most Mf taken up by anopheline vectors do not survive to become infective larvae (L3) [34,35].

CFA and anti-filarial antibody rates fell more rapidly after MDA in children <11 years of age than in the total study population; this may be because infection intensities and years of infection/exposure tend to be lower in young children than in older individuals. Although infection rates decreased in children after MDA, many young children had positive CFA and/or anti-filarial antibody tests after 3 rounds of MDA. Of course, these children had been exposed to the parasite for years prior to MDA. Children born after LF transmission has been interrupted should not have positive CFA or antibody tests [30]. Surveillance activities to verify interruption of transmission should focus on testing young children. Mosquito monitoring provides a non-invasive means of detecting residual infections in communities if the number of young children available for testing is small.

This study provided interesting longitudinal data on effects of MDA on Mf clearance in individuals and on incidence rates for different filariasis parameters. Mf clearance rates in this study after one or more annual doses of DEC with albendazole were higher than those reported from clinical trials performed in Sri Lanka and Egypt [36–38]. However, while all subjects in the clinical trials had high baseline Mf counts, all Mf carriers were considered in current community-based study. The current study also found that Mf clearance rates after MDA were lower in persons with high baseline Mf counts. The incidence data are very exciting, because

they demonstrated that MDA significantly decreased the incidence of Mf, CFA, and antifilarial antibodies in the study population. This is the first study that has documented decreased filariasis incidence rates following MDA. However, incidence events observed after MDA-3 suggest that 3 rounds of MDA was not sufficient to completely eliminate LF transmission in this setting.

The impact of three rounds of MDA in the current study was at least as impressive as that recently reported from Egypt (Giza governorate) with the same MDA regimen, although this "high prevalence" study area in Egypt had lower LF infection rates before MDA (11.5% Mf, 19.0% CFA, and 3.07% mosquito DNA) and higher MDA compliance rates than our study site in Papua New Guinea [5]. This suggests that the encouraging results reported from Egypt can be replicated in areas with very different epidemiological parameters. Changes in infection parameters following MDA must be considered in the context of the local mosquito vector. An. punctulatus is a less efficient LF vector than Cx. pipiens (the principal LF vector species in Egypt). We do not know the minimum requirements for sustained transmission of W. bancrofti by An. punctulatus. However, Tisch et al recently reported that LF parameters continued to decrease in villages in a different area of Papua New Guinea (in East Sepik Province, approximately 300 km from the Usino study site) for at least 5 years after Mf rates had been reduced to low levels by 5 rounds of MDA with DEC and ivermectin [32]. It is possible that three rounds of DEC with albendazole (which reduced the Mf rate to 1.3% with a 97.9% decline in CMFL) would have been sufficient to reduce LF transmission rates to unsustainable levels in the Usino study area. The study villages received a fourth round of MDA in 2006; long term follow-up studies will be needed to determine whether four rounds of MDA have interrupted LF transmission in this area. For the time being, the results from Usino are quite encouraging. Taken together with earlier studies, they suggest that LF elimination should be feasible in Papua New Guinea and other endemic areas with Anopheles transmission if MDA can be effectively delivered to endemic populations. Prospects for LF elimination should be even brighter if MDA can be integrated with distribution of insecticide-treated bednets [11].

#### Acknowledgments

The authors would like to thank the field teams (for community relations, MDA, and specimen collection) and laboratory staff (for entomology, parasitology, and serology work) of the Papua New Guinea Institute of Medical Research in Madang for diligent effort needed for this project. The project also benefited from administrative support provided by PNG IMR staff in Madang and Goroka. Special thanks go to Nandao Tarongka and Moses Biasor who led the field research teams and to Absalom Mai and Florence Bockarie who performed the antibody and antigen assays.

#### **Author Contributions**

Conceived and designed the experiments: GJW WK SJL SAW CLK JWK MJB. Performed the experiments: GJW WK MS SJL CLK MJB. Analyzed

#### References

- Ottesen EA, Duke BO, Karam M, Behbehani K (1997) Strategies and tools for the control/elimination of lymphatic filariasis. Bull World Health Org 75: 491–503.
- Ottesen EA (2006) Lymphatic filariasis: Treatment, control, and elimination. Adv Parasitol 61: 395–441.
- World Health Organization (2008) Global programme to eliminate lymphatic filariasis: Progress report on mass drug administration in 2007. Wkly Epidemiol Rec 83: 333–341.
- El Setouhy M, Abd Elaziz KM, Helmy H, Farid HA, Kamal HA, et al. (2007) The effect of compliance on the impact of mass drug administration for elimination of lymphatic filariasis in Egypt. Am J Trop Med Hyg 77: 1069–1073.
- Ramzy RMR, El Setouhy M, Helmy H, Ahmed ÉS, Abd Élaziz KM, et al. (2006) Effect of yearly mass drug administration with diethycarbamazine and albendazole on bancrofitian filariasis in Egypt: a comprehensive assessment. Lancet 367: 992–999.
- Bockarie MJ, Tisch DJ, Kastens W, Alexander ND, Dimber Z, et al. (2002) Mass treatment to eliminate filariasis in Papua New Guinea. N Engl J Med 347: 1841–1848.
- Michael E, Bundy DAP (1997) Global mapping of lymphatic filariasis. Parasitology Today 13: 472–476.
- Bryan JH (1986) Vectors of Wuchereria bancrofti in the Sepik Provinces of Papua New Guinea. Trans R Soc Trop Med Hyg 80: 123–131.
- Ichimori K, Crump A (2005) Pacific collaboration to eliminate lymphatic filariasis. Trends Parasitol 21: 441–444.
- Webber RH (1975) Vector control of filariasis in the Solomon Islands. Southeast Asian J Trop Med Public Health 6: 430–434.
- Bockarie MJ, Kazura JW (2003) Lymphatic filariasis in Papua New Guinea: prospects for elimination. Med Microbiol Immunol (Berl) 192: 9–14.
- Bockarie MJ, Tavul L, Ibam I, Kastens W, Hazlett F, et al. (2007) Efficacy of single-dose diethylcarbamazine compared with diethycarbamazine combined with albendazole against *Wuchereria bancrofti* infection in Papua New Guinea. Am J Trop Med Hyg 76: 62–66.
- Selve BP, Bwadua S, Misa M, James K, Usurup JP, et al. (2000) Community empowerment in the control of lymphatic filariasis in Misima, Milne Bay Province using diethylcarbamazine in combination with albendazole. PNG MedJ 43: 183–187.
- World Health Organization (2000) Preparing and implementing a national plan to eliminate lymphatic filariasis in countries where onchocerciasis is not coendemic. Geneva: WHO.
- Charlwood JD, Bryan JH (1987) A mark-recapture experiment with the filariasis vector *Anopheles punctulatus* in Papua New Guinea. Ann Trop Med Parasitol 81: 429–436.
- Desowitz RS, Hitchcock JC (1974) Hyperendemic bancroftian filariasis in the Kingdom of Tonga: the application of the membrane filter concentration technique to an age-stratified blood survey. Am J Trop Med Hyg 23: 877–879.
- Remme J, Ba O, Dadzie K, Karam M (1986) A force-of-infection model for onchocerciasis and its applications in the epidemiological evaluation of the Onchocerciasis Control Programme in the Volta River basin area. Bulletin of the World Health Organization 64: 667–681.
- Chandrasena TG, Premaratna R, Abeyewickrema W, de Silva NR (2002) Evaluation of the ICT whole-blood antigen card test to detect infection due to *Wuchereria bancrofti* in Sri Lanka. Trans R Soc Trop Med Hyg 96: 60–63.
- Weil GJ, Lammie PJ, Weiss N (1997) The ICT Filariasis Test: A rapid-format antigen test for diagnosis of bancroftian filariasis. Parasitol Today 13: 401–404.
- Ramzy RM, Helmy H, Faris R, Gad AM, Chandrashekar R, et al. (1995) Evaluation of a recombinant antigen-based antibody assay for diagnosis of bancroftian filariasis in Egypt. Ann Trop Med Parasitol 89: 443–446.
- Chandrashekar R, Curtis KC, Ramzy RM, Liftis F, Li B-W, et al. (1994) Molecular cloning of *Brugia malayi* antigens for diagnosis of lymphatic filariasis. Mol Biochem Parasitol 64: 261–274.

the data: GJW WK MS SJL SAW MJB. Contributed reagents/materials/ analysis tools: GJW SJL SAW. Wrote the paper: GJW WK SJL CLK JWK MJB.

- Lammie P, Weil G, Rahmah N, Kaliraj P, Steel C, et al. (2004) Recombinant antigen based assays for the diagnosis and surveillance of lymphatic filariasis - a multicenter trial. Filaria Journal 3: 9.
- Plichart C, Laney SJ, Sechan Y, Davies N, Legrand AM (2007) Correction: PCR and dissection as tools to monitor filarial infection of *Aedes polynesiensis* mosquitoes in French Polynesia. Filaria J 6: 5.
- 24. Plichart C, Sechan Y, Davies N, Legrand AM (2006) PCR and dissection as tools to monitor filarial infection of *Aedes polynesiensis* mosquitoes in French Polynesia. Filaria J 5: 2.
- 25. Williams S, Laney S, Bierwert L, Saunders L, Boakye D, et al. (2002) Development and standardization of a rapid PCR-based method for the detection of *Wuchereria bancrofti* in mosquitoes for xenomonitoring the human prevalence of bancroftian filariasis. Ann Trop Med Parasitol 96: S41–46.
- Laney SJ, Buttaro CJ, Visconti S, Pilotte N, Ramzy RMR, et al. (2008) A reverse transcriptase-PCR assay for detecting filarial infective larvae in mosquitoes. PLoS Negl Trop Dis 2: e251. doi:10.1371/journal.pntd.0000251.
- Katholi CR, Toé L, Merriweather A, Unnasch TR (1995) Determining the prevalence of *Onchocerca volvulus* infection in vector populations by polymerase chain reaction screening of pools of black flies. J Infect Dis 172: 1414–1417.
- Njenga SM, Wamae CN, Njomo DW, Mwandawiro CS, Molyneux DH (2008) Impact of two rounds of mass treatment with diethycarbamazine plus albendazole on *Wuchereria bancrofti* infection and the sensitivity of immunochromatography test in Malindi, Kenya. Trans R Soc Trop Med Hyg: in press.
- Weil GJ, Ramzy RM, El Setouhy M, Kandil AM, Ahmed ES, et al. (1999) A longitudinal study of Bancroftian filariasis in the Nile Delta of Egypt: baseline data and one-year follow-up. Am J Trop Med Hyg 61: 53–58.
- Weil GJ, Ramzy RMR (2007) Diagnostic tools for filariasis elimination programmes. Trends Parasitol 23: 78–82.
- Simonsen PE, Magesa SM, Meyrowitsch DW, Malacela-Lazaro MN, Rwegoshora RT, et al. (2005) The effect of eight half-yearly single-dose treatments with DEC on *Wuchereria bancrofti* circulating antigenemia. Trans R Soc Trop Med Hyg 99: 541–547.
- Tisch DJ, Bockarie MJ, Dimber Z, Kiniboro B, Tarongka N, et al. (2008) Mass drug administration trial to eliminate lymphatic filariasis in Papua New Guinea: Changes in microfilaremia, filarial antigen, and Bm14 antibody after cessation. Am J Trop Med Hyg 78: 289–293.
- 33. Fischer P, Erickson SM, Fischer K, Fuchs JF, Rao RU, et al. (2007) Persistence of *Brugia malayi* DNA in vector and non-vector mosquitoes: implications for xenomonitoring and transmission monitoring of lymphatic filariasis. Am J Trop Med Hyg 76: 502–507.
- McGreevy PB, Bryan JH, Oothuman P, Kolstrup N (1978) The lethal effects of the cibarial and pharyngeal armatures of mosquitoes on microfilariae. Trans R Soc Trop Med Hyg 72: 361–368.
- Pichon G (2002) Limitation and facilitation in the vectors and other aspects of the dynamics of filarial transmission: the need for vector control against *Anopheles*-transmitted filariasis. Annals of Tropical Medicine and Parasitology 96: S143–S152.
- 36. El Setouhy M, Ramzy R, Ahmed E, Kandil A, Hussain O, et al. (2004) A randomized clinical trial comparing single- and multi-dose combination therapy with diethylcarbamazine and albendazole for treatment of bancroftian filariasis. Am J Trop Med Hyg 70: 191–196.
- Helmy H, Weil GJ, Ellethy AS, Ahmed ES, El Setouhy M, et al. (2006) Bancroftian filariasis: Effect of repeated treatment with diethycarbamazine and albendazole on microfilaremia, antigenemia, and anti-filarial antibodies. Trans R Soc Trop Med Hyg 100: 656–662.
- Ismail MM, Jayakody RL, Weil GJ, Fernando D, deSilva MSG, et al. (2001) Long-term efficacy of single-dose combinations of albendazole, ivermectin, and diethylcarbamazine for the treatment of lymphatic filariasis. Trans Roy Soc Trop Med Hyg 95: 332–335.